Author’s response to reviews

Title: Antibacterial, Antioxidant and Immuno-modulatory Properties in extracts of Barleria Lupulina Lindl.

Authors:
RESHMA KUMARI (reshmugupta25@gmail.com)
Sumit Kumar (sumitarya360@gmail.com)
Ashwani Kumar (ashwanipharma03@gmail.com)
Kapil Goel (kapilgoel9@gmail.com)
Ramesh Dubey (profrcdubey@gmail.com)

Version: 4 Date: 17 Oct 2017

Author’s response to reviews:

Reviewer's answer

1. Some raw data has been also published regarding this research which has been discussed earlier and their references have been included in this manuscript.

2. Done as per your suggestion.